Author's response to reviews

Title: Using mobile technology to support lower-salt food choices for people with cardiovascular disease: protocol for the SaltSwitch randomized controlled trial

Authors:

Helen Eyles (h.eyles@nihi.auckland.ac.nz)
Rebecca McLean (r.mclean@auckland.ac.nz)
Bruce Neal (bneal@georgeinstitute.org.au)
Robert N Doughty (r.doughty@auckland.ac.nz)
Yannan Jiang (y.jiang@auckland.ac.nz)
Cliona Ni Mhurchu (c.nimhurchu@auckland.ac.nz)

Version: 2  Date: 7 September 2014

Author's response to reviews:

To the editors,

As requested, we have now added the date that the trial was registered alongside the trial registration number.

Yours sincerely
Helen Eyles (corresponding author)